Author's response to reviews

Title: Performance of Commercial Dengue NS1 ELISA and Molecular Analysis of NS1 Gene of Dengue Viruses Obtained during Dengue Surveillance in Indonesia

Authors:

A Aryati (dr_aryati@yahoo.com)
Hidayat Trimarsanto (anto@eijkman.go.id)
Benediktus Yohan (yohan@eijkman.go.id)
Puspa Wardhani (puspa_pk@yahoo.co.id)
Sukmal Fahri (sukmalfahri@yahoo.co.id)
R. Tedjo Sasmono (sasmono@eijkman.go.id)

Version: 2 Date: 30 August 2013

Author's response to reviews:

Dear Editor,

We have revised our manuscript according to initial assessment, to include the following:
- Competing Interests and
- Authors Contribution

A new version of the manuscript is uploaded.

Yours sincerely,
Tedjo Sasmono